Author’s response to reviews

Title: Support and Assessment for Fall Emergency Referrals (SAFER 1) trial protocol Full title: Computerised on-scene decision support for emergency ambulance staff to assess and plan care for older people who have fallen: evaluation of costs and benefits using pragmatic cluster randomised trial

Authors:

Helen Snooks (h.a.snooks@swansea.ac.uk)
Wai-Yee Cheung (w.y.cheung@swansea.ac.uk)
Jacqueline Close (j.close@powmri.edu.au)
Jeremy Dale (jeremy.dale@warwick.ac.uk)
Sarah Gaze (s.a.gaze@swansea.ac.uk)
Ioan Humphreys (i.humphreys@swansea.ac.uk)
Ronan Lyons (r.a.lyons@swansea.ac.uk)
Suzanne Mason (s.mason@sheffield.ac.uk)
Yasmin Merali (yasmin.merali@wbs.ac.uk)
Julie Peconi (j.peconi@swansea.ac.uk)
Ceri Phillips (c.j.phillips@swansea.ac.uk)
Judith Phillips (j.peek.phillips@swansea.ac.uk)
Stephen Roberts (stephen.e.roberts@swansea.ac.uk)
Ian Russell (i.t.russell@swansea.ac.uk)
Antonio Sánchez (a.sanchez@swansea.ac.uk)
Mushtaq Wani (mushtaq.wani@swansea-tr.wales.nhs.uk)
Bridget Wells (b.wells@swansea.ac.uk)
Richard Whitfield (richard.whitfield@ambulance.wales.nhs.uk)

Version: 2 Date: 19 July 2009

Author’s response to reviews:

Unfortunately a clerical error meant that the wrong version of our manuscript was submitted. This mistake concerns the study outcomes only- we had further refined our primary and principal outcomes which are now correct in this version. Please accept our apologies for any inconvenience this causes you.